Author's response to reviews

Title: Use of Cerebral State Index to Predict Long-Term Unconsciousness in Patients after Elective Craniotomy with Delay Recovery

Authors:

Ming Xu (sunnykiwi2008@163.com)
Yan-Ni Lei (leiyanni1981@163.com)
Jian-Xin Zhou (zhoujx.cn@gmail.com)

Version: 2 Date: 10 March 2010

Author's response to reviews:

Dear Editor:

On behalf of my co-authors, I am submitting the manuscript entitled “Use of cerebral state index to predict long-term unconsciousness in patients after elective craniotomy with delay recovery” for possible publication in BMC Neurology. This research is original, and has not been published in whole or in part elsewhere, nor has it been simultaneously submitted to any other journal. I attest to the fact that all authors listed on the title page have contributed to preparing the manuscript, attest to the validity and legitimacy of the data and its interpretation, and agree to its submission to BMC Neurology.

Sincerely yours

Jian-Xin Zhou, MD
Professor and Chairman
Department of Critical Care Medicine
Beijing Tiantan Hospital, Capital Medical University, Beijing 100050, China
Telephone: 8610 67098019
Fax: 8610 67096579
E-mail: zhoujx.cn@gmail.com